Author’s response to reviews

Title: Systematic analysis of the clinical and biochemical characteristics of maternally inherited hypertension in Chinese Han families associated with mitochondrial genome mutations

Authors:

yuqi liu (ametufo980869@163.com)
chao zhu (zhuchao@163.com)
jie yang (yangjie@163.com)
tong yin (yintong@163.com)
jinliao gao (gaojinl@163.com)
zongbin li (lizongb@163.com)
qinhua ma (magingh@163.com)
yang li (liyang@163.com)
minxin guan (guanmx@163.com)
yundai chen (ametufo4290@gmail.com)

Version: 2 Date: 24 May 2014

Author’s response to reviews: see over
Dear editor,

On behalf of my co-authors, thank you to receive the manuscript titled “Systematic analysis of the clinical and biochemical characteristics of maternally inherited hypertension in Chinese Han families associated with mitochondrial genome mutations” submitted for publication in your magazine.

We had revised the article according to your requirements:

1. Title page: email addresses for all authors.
   Had revised the title page involved the email addresses of all authors and the full postal address of the submitting author. (page 1)

2. Line Numbering.
   Had inserted the line and page numbers.

3. Competing Interests.
   Had inserted the ‘Competing interests’ section. (page 2)

4. Authors’ Contributions.
   Had inserted the Contributions section. (page 11)

5. Acknowledgements.
   Had inserted the Acknowledgements section. (page 11)

Sincerely yours,

Address : Institute of Geriatric Cardiology, Chinese PLA General Hospital,
          Road Fuxing, Haidian District, Beijing, China, Zip 100853

E-mail: ametuofo980869@163.com

Tel:  86-10-66936762
FAX: 86-10-88270497